Author's response to reviews

Title: The effect of cognitive-motor dual-task training on cognitive function and plasma amyloid beta peptide 42/40 ratio in healthy elderly persons: a randomized controlled trial

Authors:

Hisayo Yokoyama (yokoyama@sports.osaka-cu.ac.jp)
Okazaki Kazunobu (okazaki@sports.osaka-cu.ac.jp)
Daiki Imai (dimai@sports.osaka-cu.ac.jp)
Yoshihiro Yamashina (y-yamashina@pt-u.aino.ac.jp)
Ryosuke Takeda (wind-05@med.osaka-cu.ac.jp)
Nooshin Naghavi (nooshin@med.osaka-cu.ac.jp)
Akemi Ota (ota@isc.osakac.ac.jp)
Yoshikazu Hirasawa (yokkunn.h.1124@gmail.com)
Toshiaki Miyagawa (miyagawa@sports.osaka-cu.ac.jp)

Version: 2 Date: 14 October 2014

Author's response to reviews: see over
Ms. Danica Jose  
Journal Editorial Office of BioMed Central

Title of the manuscript: The effect of cognitive-motor dual-task training on cognitive function and plasma amyloid β peptide 42/40 ratio in healthy elderly persons: a randomized controlled trial

Authors: Hisayo Yokoyama, Yoshihiro Yamashina, Ryosuke Takeda, Nooshin Naghavi, Akemi Ota, Yoshikazu Hirasawa, Daiki Imai, Kazunobu Okazaki, and Toshiaki Miyagawa

Dear Ms. Danica Jose

We appreciate you to kindly notify us the points of the manuscript we need to revise. According to your assignment, we added the essential data regarding 95% confidence intervals to the paragraph of The outcome in cognitive function (p14. line 16～) in the RESULTS section and the corresponding parts in the Abstract and the Figure legends. We also added some revisions to the paragraph of Statistical Analysis (p12. line 10～) in the METHODS section. Please confirm the revised manuscript.

Thank you for your time and consideration.

Sincerely yours,

Hisayo Yokoyama
Department of Environmental Physiology for Exercise
Osaka City University Graduate School of Medicine
3-3-138, Sugimoto, Sumiyoshi-ku, Osaka, 545-8585, Japan
Tel.& Fax: (Japan code 81) 6-6605-2947
E-mail: yokoyama@sports.osaka-cu.ac.jp